Author's response to reviews

Title: Acute febrile neutrophilic dermatosis (Sweet's syndrome) in a child, associated with a Rota virus infection: a report of an atypical case

Authors:

   Alexandros makis (amakis@cc.uoi.gr)
   Stavros Stavrou (sstavrou77@gmail.com)
   Nikolaos Chaliasos (nhalias@cc.uoi.gr)
   Aikaterini Zioga (aikaterini.zioga@medai.gr)
   Antonios P Vlahos (anvlahos@uoi.gr)
   Georgios Gaitanis (ggaitan@cc.uoi.gr)
   Antigone Siamopoulou (asiamop@cc.uoi.gr)
   Ioannis D Bassukas (ibassuka@cc.uoi.gr)

Version: 2 Date: 26 April 2010

Author's response to reviews:

We would like to thank the editorial team for accepting the manuscript and the referees for the extremely helpful comments.

Response to reviewers:
Referee 1: -
Referee 2:
1. The term “atypical” has been added in the title and in the first sentence of the Discussion
2. The differential diagnosis to acute bullous pyoderma gangrenosum has been included in the Case Presentation section

Editors:
1. The ethnicity of the patient has been included
2. The last paragraph of the Discussion is the conclusion section